cos423: Theory of Algorithms Handout #3 Michael Goldwasser Princeton University Tuesday, February 10, 1998

Office Hours and Precepts

Email Questions

For those wishing to ask question through email, we encourage you to send email to cos423@links.princeton.edu, which will be read by all three of us. This is likely to get you the quickest response as any one of us may be around to read your question. Of course, if you have a question for a specific individual, such as the TA who graded your last homework, then please email that person directly.

Newsgroup Questions

If you have a question of general class interest, you may post the question to the class newsgroup (pu.cs.423). However please be careful not to post anything that might give other students “unfair” help in doing their own homework.

Office Hours

ffl Instructor: Michael Goldwasser

e-mail: wass@cs.princeton.edu office: Room 204 (CS Building) telephone: 258-1749 office hours: Tu/Th 10:30–11:30am, W 2–3pm, (or by appointment)

ffl Teaching Assistant: Amit Chakrabarti e-mail: amitc@cs.princeton.edu office: Room 313 (CS Building) telephone: 258-6126 office hours: Mon. 3–5pm

ffl Teaching Assistant: Kostas Tsioutsiouliklis e-mail: kt@cs.princeton.edu office: Room 317 (CS Building) telephone: 258-1796 office hours: Fri. 4–6pm

Precept

For now, we have chosen to hold a single weekly precept. If we have enough interest for those with scheduling conflicts, we may add a second option later in the semester.

ffl Precept Section: Amit Chakrabarti time: Wednesday, 8:00-9:00pm place: Room 105 (CS Building)